Author's response to reviews

Title: Who pays and who benefits? How different models of shared responsibilities between formal and informal carers influence projections of costs of dementia management.

Authors:

Victor Vickland (victor.vickland@unsw.edu.au)
Joel Werner (j.werner@unsw.edu.au)
Thomas Morris (thomas.morris@unsw.edu.au)
Geoff McDonnell (gmcdonne@bigpond.net.au)
Brian Draper (b.draper@unsw.edu.au)
Lee-Fay Low (lf.low@unsw.edu.au)
Henry Brodaty (h.brodaty@unsw.edu.au)

Version: 5 Date: 5 October 2011

Author's response to reviews: see over
Cover letter:

Major revisions (we require the author to make these changes)

Authors' contributions: Please indicate that all authors read and approved the final manuscript.

All authors have read and approved the manuscript. This has been made explicit in the acknowledgements section.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Comments: Please remove post-it style comments. Eg: Insert table 1 here.

This has been completed

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

In correspondence with Ms Robyn Hughes it was noted that the attached pdf figures can be edited by you.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

This has been completed

Thank you

Other changes (as outlined in the previous cover letter uploaded on the 28th September 2011):

In correspondence with the authors of the article, the following slight changes were made:

- The title was reworded
- Certain text was altered for grammar, spelling and composition
- Certain areas have been elaborated upon to provide clarification.